Author’s response to reviews

Title: Prevalence of frailty and contributory factors in three Chinese populations with different socioeconomic and healthcare characteristics

Authors:

Jean Woo (jeanwoowong@cuhk.edu.hk)
Zheng Zheng (zzheng25@hotmail.com)
Jason Leung (jason-leung@cuhk.edu.hk)
Piu Chan (pbchan@hotmail.com)

Version: 3 Date: 27 Nov 2015

Author’s response to reviews:

Indicates author’s reply

1. Please provide a clear statement of ethical approval from a named ethics committee. Please also provide a separate statement detailing the informed consent obtained for the participants, specifying whether the consent obtained was written or verbal.

- The following statements are added under the Participants

Ethical approval was obtained from the Research Ethics Committee of Xuanwu Hospital of the Capital Medical University.

Ethical approval was obtained from the Clinical Research Ethics Committee of the Chinese University of Hong Kong and Hospital Authority NTE Cluster.

Written informed consents were obtained from all participants in both sites.